Author's response to reviews

Title: The study protocol of a cluster-randomised controlled trial of family-mediated personalised activities for nursing home residents with dementia

Authors:

Eva S van der Ploeg (Eva.vanderPloeg@monash.edu)
Cameron J Camp (Cameron@cen4ard.com)
Barbara Eppingstall (Barbara.Eppingstall@monash.edu)
Susannah J Runci (Susannah.Runci@monash.edu)
Daniel W O'Connor (Daniel.OConnor@monash.edu)

Version: 3 Date: 10 January 2012

Author's response to reviews:

We have made the following formatting changes:
- We have changed the heading into: Abstract; Background; Methods/Design; Discussion; Competing interests; Authors’ contributions; Acknowledgements (optional); References
- We have renamed Figure 1 as a table
- We have changed to formatting of e-mail addresses to: AB: abcd@institution.ac.uk
- We have referred to reference 9 in the text and updated the numbers of all following references